REVIEWS

A Reassessment of Materials Issues in Microelectromechanical Systems (MEMS)

Liudi Jiang AND S. Mark Spearing

Abstract | Over the past 7 years there has been an explosion of research activity into materials for MicroElectroMechanical Systems (MEMS). This paper reviews the current issues associated with materials for MEMS. Five topical areas are addressed: the effect of lengthscale, the selection of materials and processes, the MEMS material set, microfabrication processes and material characterization. Each of these areas is examined, with particular emphasis on the potential impact of materials solutions. The paper concludes with an assessment of the progress in MEMS materials made since 2000.

1. Introduction

MicroElectroMechanical Systems (MEMS) represent a significant industry sector, with a net value of chip level devices (i.e. unpackaged) estimated to be in the range of US\$7-\$10bn for 2006 with a sustained growth rate of 15-20% per annum over the past decade. The basis of this commercial impact is that the efficiencies of high volume production and low unit cost routinely achieved by the microelectronics industry can be translated to devices in which mechanical and electrical components are integrated within a single silicon chip (or equivalent structure). In addition to the economic benefits, unique capabilities can be achieved by such integration to realize a wide range of systems including: sensors^{1,2}, actuators³, power producing devices⁴, chemical reactors⁵, biomedical devices⁶ and for tissue engineering⁷. Furthermore the ability to combine the sensing/actuating (or other) function together with the electronics required for control and power conditioning in a single device allows for consideration of concepts such as the highly distributed networks⁸ required for health monitoring of large structures and systems⁹ or for distributed flow control¹⁰.

The continued success of MEMS depends crucially on the solution of materials issues associated with the design and fabrication of complex MEMS devices. The small scales of MEMS offers the opportunity to exploit materials which would not normally be available for large scale devices as well as taking advantage of the favourable scaling of some properties, notably fracture strength. MEMS also offer the opportunity to materials scientists and engineers to be able to characterize materials in ways that have not hitherto been possible. One of us (SMS) first examined the interactions between developments in Materials and MEMS in a review article published in 2000^{11} . In the present article we reassess the topic and the very significant developments that have occurred over the intervening seven years. Given the rate of the growth of the field of MEMS and the role of materials within it, the present article makes no pretence at being comprehensive. Interested readers are referred to several excellent broader references/reviews of MEMS technology¹²⁻¹⁴ and microfabrication¹⁵.

The structure of the remainder of the paper is as follows: Section 2 discusses the effects of length scale

Engineering Materials Research Group, School of Engineering Sciences, University of Southampton, Highfield, Southampton, SO17 1BJ, United Kingdom

Keywords:MicroElectromechanical Systems, Materials Selection, Scaling, Microfabrication, Mechanical Properties Figure 1: Scanning electron micrograph of a single *E. coli* bacterium on an antibody-coated silicon nitride cantilever oscillator¹⁸.



on MEMS design. Section 3 discusses approaches to material and process selection for MEMS. Section 4 presents the MEMS materials set. Section 4 discusses materials issues associated with key fabrication steps. Section 5 presents the key microfabrication processes used for MEMS. Section 6 presents a review of mechanical characterization approaches for MEMS and section 7 offers concluding remarks.

2. Scale issues in MEMS

The small scales associated with MEMS have proven to be beneficial to enable new functions and/or permit significant cost-reductions. For example mechanical resonators are useful for wireless communications¹⁶ and offer the potential for ultrasensitive sensing such as single-molecule detection applications^{17,18}. Fig. 1 shows a single bacterium attached to a cantilever resonator. To achieve resonant frequencies in the high MHz

Table 1: Quasi-fundamental scaling of physical parameters for MEMS.

Physical parameters	Scaling	Units	
Length	L	m	
Area	L^2	m ²	
Volume	L ³	m ³	
Surface area/volume ratio	L^{-1}	m^{-1}	
Mass	L ³	kg	
Strength	L^2	N. m ⁻²	
Inertial force	L ³	Ν	
Electrostatic force	L ²	Ν	
Piezoelectric force	L^2	Ν	
Magnetic force (electromagnet)	L^4	N	
Natural frequency	L^{-1}	Hz	
Ohmic current	L^2	А	
Resistance	L^{-1}	Ω	
Voltage	L	V	
Thermal conductance	L	$W.K^{-1}$	

or even GHz ranges with high quality factors (Q), micro/nano-scale resonators fabricated from low-loss materials such as Si¹⁹, SiN²⁰, SiC²¹ and even carbon nanotubes²², are required. Other applications requiring specific micron or even nanometre scales to ensure their functionality include; ink-jet print heads, cantilever tips for AFM and thin film magnetic disk heads for massmemory storage. By contrast in many cases, the realisation of significant cost-reduction has been a primary motivation for MEMS. Key examples include micromachined accelerometers and pressure sensors for automotive applications. Fabrication of micro-accelerometers through micromachining techniques has enabled their mass production at very low cost. These fully packaged microdevices retail for only a few US dollars each²³. Regardless of whichever is the stronger driver for creating small dimensioned devices, the effect of scale is important in all MEMS as it affects system design, material properties and manufacturing processes. This section highlights the main scale effects in MEMS and resulting issues attributable to them.

2.1. Scaling Laws

At the macro scale, fundamental scaling laws are encountered in which physical constants and material properties are independent of scale. These may cease to be appropriate as the scale diminishes. More properly such scaling laws might be termed "quasi-fundamental"¹¹. Inevitably they break down at some scale, and are not truly fundamental or universal in their applicability. Cube-square scaling is one of the most important such scaling laws for devices in which performance is governed by the ratio between parameters with volumetric and areal dependencies respectively, for instance the shock resistance of MEMS sensors or the power/volume of power generators. Table 1 lists various physical parameters and their quasi-fundamental scale dependencies that are relevant to MEMS designs. Material properties become scale dependent when the length scale of the structure being characterized approaches the length scale of the mechanism governing the property of interest at smaller scales. For example, the plastic response of metals is due to the mechanism of dislocation motion. Thin metallic films deposited on substrates are stronger than the bulk material²⁴. Dislocation formation and motion in thin films with thickness below a characteristic thickness are restricted and consequently high mechanical strengths can occur. The increase in mechanical strength also allows higher residual strengths to be exhibited than in macro-scale structures, which is a key challenge for MEMS. Obviously, these scaling effects in

Figure 2: Two-chip accelerometer in an industry standard 16-pin dual-in line package (DIP)²⁸.



material properties play roles in defining device performance and therefore should be taken into consideration during MEMS design. Meanwhile, it is also important to recognise the limitation of current fabrication techniques which often restrict the geometries of the key elements in MEMS. In particular, the dimensional tolerances in various lithography techniques very much determine the minimum feature size of MEMS, while element shapes are mainly attributed to the deposition and etching techniques used. These dependencies result in indirect scaling, which is not attributable to a single mechanism or property.

2.2. Implications of small scale

The shrinkage of devices into small scales not only generates many advantageous material properties but also complications in many aspects of MEMS. One of the primary consequences is the increased surface area to volume ratio at small scales which has a negative effect in many MEMS applications especially microfluidic channels²⁵. The increased influence of surface tension and viscous losses in microfluidic channels limits the potential for down sizing in these systems. In the same vein, the performance of MEMS inertial sensors and other force sensors is ultimately limited by thermal noise²⁶. Small structures also have limited thermal stability which has become a critical bottleneck in applications such as MEMS actuators in magnetic hard disk drives where a stable actuator head must follow the narrow data track with high accuracy²⁷. Finally, with the rapid emergence of functional MEMS prototypes, in order to develop these prototypes into commercialized products, packaging has become a prominent factor influencing the

overall product dimensions as well as the unit cost of MEMS. Many MEMS especially those containing moving components require hermetic or nearhermetic packaging due to their susceptibility to environmental damage such as damping and moisture. MEMS' relatively high surface-area-tovolume ratio makes the hermetic environment critical. Fig. 2 shows a typical accelerometer package²⁸. This induces significant challenges on the dimensional reduction of final MEMS products. Similarly RF MEMS packaging consumes a large area of the devices^{29,30} and often reaches more than 70% of the total cost of the devices³¹. Packaging is a major factor contributing to the failure of many MEMS. Therefore, in order to realize the performance and dimension advantages of MEMS products, it is necessary to develop cost-effective and reliable packaging with small dimensions.

2.3. The potential for NEMS?

In recent years, a multitude of research has been carried out to scale MEMS to submicron dimensions thereby creating Nanoelectromechanical Systems (NEMS) with a view to achieving advanced device performance as well as exploiting attractive material properties occurring at nanometer scales. The anomalous properties of carbon nanotubes (CNT) are perhaps one of the best examples of substantially enhanced material properties at the nano-meter scale. Since the phonon mean free path is comparable to the length of CNTs, ballistic thermal and electrical conduction occur in CNTs replacing the mechanisms described by Fourier's law in bulk materials^{32,33}. By virtue of their graphene structure CNTs also show extremely high Young's moduli (on the order of 1TPa)³⁴. These electrical and mechanical properties suggest CNTs might be excellent building blocks for NEMS. As a result, CNT elements have been explored for resonators³⁵, sensors^{36,37} and relays³⁸. Solely realising a dimensional advantage is the primary drive for many NEMS developments. For instance, due to the nanoscale track widths in magnetic recording, magnetic recording heads with nanometer precision are required. Prototypes of heads with trackwidths smaller than 100 nm have been achieved enabling an areal density of the order of Tbit/in² ³⁸. The continued reduction in scales imposes increasing challenges on device design, including the lack of information of material properties at the nanometre scale. There are also significant challenges to the fabrication and testing of electromechanical devices at the nanometer scale. For example, a resonant frequency of over 1GHz has only been achieved at low temperatures (4.2 K) in ultrahigh vacuum³⁹. This inevitably limits many NEMS to specific applications, which could offer satisfactory working environments.

3. Materials and Process Selection

Hitherto commercial MEMS products have largely utilized the materials and process set bequeathed by the semiconductor microelectronics industry. This has had the consequence that the devices which have been realized have tended to be those which can be made using the available materials and processes. As the MEMS sector expands and diversifies it is increasingly important to select materials and processes that are optimized with respect to the functional requirements of the devices themselves. To this end it is worthwhile examining the selection of materials and processes that are available to the MEMS designer using quantitative performance metrics by which to compare the candidates.

3.1. Material Selection

The fundamental approach to such selection activities has been developed by Ashby⁴⁰ and has subsequently been adapted to MEMS^{11, 41,42}. At the heart of this approach to materials selection is the idea that the "performance" of a mechanical design can be expressed in terms of the functional requirements, geometric parameters, and material indices. A materials index is a combination of material properties that govern the scaling of the design performance. For instance the resonant frequency of a device scales with $\sqrt{(E/\rho)}$, the maximum acceleration to fracture a sensor scales with (σ_f/ρ) and the deflection capability of a flexure scales with (σ_f / E), where E is the Young's modulus, ρ the density and σ_f the failure strength of the material. Such metrics are easy to derive and a convenient way of presenting the data is

10⁴ ms⁻¹ Diamond 1000 AI C SiC ٠w Young's Modulus, E (GPa) Śi N, Pt Quartz 100 Au GaAs AI SiO. 10 3 x 10³ ms⁻¹ $(\dot{E}/\rho)^{1/2}$ 10³ ms⁻¹ Polymers & Elastomers 2 6 8 10 20 4 Density, ρ (Mg/m³)

Figure 3: MEMS Materials plotted on an Ashby-style Material Selection Chart for Young's Modulus and Density.

to construct "charts" in which the properties of engineering materials are plotted on axes of the relevant material properties for a particular application. The performance indices can then be plotted as functions on the same axes. For more complicated designs, where multi-objective is required other techniques can be employed⁴³. For the purposes of illustration it is instructive to consider the cases in which the information can be presented on a materials selection chart.

Figure 3 shows an example of the case for Young's modulus and density. A range of materials are included, consisting of those commonly used for MEMS devices and a few others that might be considered candidates. Contours of equal $\sqrt{(E/\rho)}$ are also plotted on the same chart. Materials that offer higher resonant frequency for a given device geometry lie to the upper left of the chart. It is clear that commonly used MEMS materials such as Si and SiN_x are amongst the better materials for such an application. High stiffness ceramics such as SiC and Al₂O₃ offer the prospect of some improvement, with only diamond, or diamond like carbon offering more than a factor of two in higher frequency capability. A more detailed discussion of this selection methodology together with material property charts for a range of materials and properties is provided in reference [42].

The overall exercise described in [42] leads to the conclusion that for most existing mechanical sensor applications (pressure sensors, accelerometers and gyroscopes) Si and the other legacy CMOS materials are quite well suited to the functional requirements of the devices. Si is a reasonably stiff material, with low density, high fracture strength and low loss coefficient. Materials such as diamond-likecarbon, with its much higher stiffness have the potential to exceed the capabilities of Si, however the cost of developing and optimizing processes and tool sets for any new material for a large volume application provides a significant barrier to entry for any new candidate material. Furthermore the ability to integrate the mechanical elements of MEMS with the electronic functionality may be compromised if non-CMOS compatible materials are introduced.

The opportunity for integrating novel materials may be broader in the realm of functional materials for actuators, where piezoelectric, shape memory and thermo-electric actuation principles can be utilized. Similar selection principles can be applied. Recent work^{44–46} on bimaterial electro thermal and piezoelectric bimaterial actuators has examined the suitability of different pairs of material for creating actuators optimized against the metrics of force, displacement, work, frequency and efficiency. The actuators consist of a cantilever beam with one material deposited on a substrate of another material. For an electrothermal actuator one or other of the two materials is heated resistively causing a thermal expansion, resulting in force or deformation. In this case the approach is to identify the capability of a candidate material on a given substrate. It is important to recognize that the optimum thickness ratio between the pair of materials will also depend on the material properties, particularly the Young's moduli. This approach allows the comparison of candidate material pairs via similar material selection charts to those described above. Figure 4 shows such a chart for a wide range of materials. The axes of the chart are the moduli and thermal expansion coefficient of the material. Contours of performance for displacement, force and work are overlayed. It is significant that the optimal materials choices vary depending on the performance metric chosen. Al/Si is found to be a very good material combination for such actuators, which is convenient given that this combination is compatible with the CMOS material set.

3.2. Process Selection

It is also instructive to compare the capabilities of the available processes. This has been achieved by surveying the literature and identifying generic "process chains" by which canonical structures, such as trenches and suspended cantilever beams or membranes can be achieved⁴⁷. Metrics such as the absolute dimensions, tolerances, surface roughness etc. have been identified for each process chain. These metrics can then be used to compare the capabilities of candidate process chains to yield a particular geometric feature. These results can be displayed graphically in the form of process selection charts in which pairs of process metrics are cross-plotted and the envelopes associated with particular process chains are superimposed. An example is given in figure 5 for the in plane and out of plane dimensions of a trench. This exercise reveals that for the most part in plane features are limited by lithography, and that relatively few processes are capable of yielding features with sub 1 micron resolution. The maximum feature depth achievable is of the order 100 μ m–1 mm, which is limited by reaction product removal for etching processes and mask uniformity microstructure control issues for deposition-based processes such as LIGA. Similar consideration of reaction kinetics and the diffusion of reaction products result in pronounced aspectratio limits for most of the process chains examined, i.e. a linear dependence of etch depth achievable on in-plane dimensions. Minimum etch depths are due to a combination of factors including the accuracy of

chemical mechanical polishing to define the surface of the initial wafer and the ability to use etch stops or timed etches or deposition to achieve a target feature depth. Although these charts are constructed based on purely empirical data culled from the literature, they do provide valuable insight as to where there might be scope for process improvement by developing new processes. Furthermore the charts covering process chain tolerances can provide valuable assessments of whether a particular device's dimensions can be controlled by the process alone or whether subsequent electrical or mechanical calibration and tuning would be required.

3.3. Sensor and Actuator Mechanism Selection

Having investigated the selection of materials and processing route it is natural to use similar approaches to investigate the selection of operating principles of MEMS sensors and actuators themselves. Clear performance metrics can be identified for sensors and actuators. These include force, displacement, frequency and the resolution achievable for each of the preceding parameters. The capabilities of fabricated MEMS transducers in the literature have been evaluated and plotted on process capability charts⁴⁸. These allow the comparison of MEMS devices and transduction principles with each other, but perhaps of more interest is the comparison of MEMS with macroscale transducers for the same purpose. This exercise sheds light on to the wide breadth of sensing and particularly actuating principles that have been attempted in microfabricated devices. Figure 6 provides an example of a chart for MEMS actuators, comparing achievable force and displacement. The MEMS are generally distinct from macroscale actuators in terms of their lower force capability, however there is less distinction in terms of the maximum displacements achievable. This particular chart clearly indicates that there is much relatively unexplored space for MEMS actuators, where hitherto electrostatic actuation with some use of piezoelectric actuation has dominated. Effective SMA actuators apparently have great potential for high force and high work (force-displacement product) actuators. Hybrid actuators such as scratch drives and external field actuators offer promise for increasing the displacement capability above that achievable by existing actuator physics. More broadly this mapping exercise provides a clear basis for selecting actuation principle, and also for identifying gaps and opportunities to create more effective MEMS transducers or to insert MEMS in place of macroscale devices.

In conclusion the quantification of performance and the selection of materials and processes is an

Properties	Si	SiO ₂	Si_3N_4	SiC	Diamond
Density (kg/m ³)	2330	2200	3300	3300	3510
Modulus (GPa)	129–187	73	304	448	1050
E/ρ (GN/kg.m)	72	36	92	130	295
Hardness (kg.mm ⁻²)	1000	710–790	1580	3500	10000
Fracture strength (MPa)	4000	1000	1000	2000	1000
Thermal conductivity at 300 K (W/cm.K)	1.5	0.014	0.3	4.9	20
Thermal expansion coefficient (10E-6.K ⁻¹)	2.6	0.4–12.3	3.3	3.8	1.1
Max. operation temperature	300	1100	1000	1240	1100
Dielectric constant	11.9	3.9	7.5	9.7	5.5

Table 2: Properties of commonly utilised MEMS materials.

important step in moving MEMS technology away from a path in which capabilities are entirely dictated by what is easily achieved using the materials, processes and transduction principles which have been inherited from Microelectronics. These exercises allow the identification of promising new materials, processes and transduction principles, as well as allowing a clear quantification of where the benefits lie with employing the existing options.

4. The MEMS Material Set

The available materials have played a key role in determining the classes of MEMS developed thus far. Table 2 lists the properties of the principal MEMS materials available. Extensively used in microelectronics industry, this Si-compatible material set has advantages for MEMS in terms of some material properties, its microfabrication feasibility and the availability of reasonably low-cost substrates. Within this material set Si is mainly utilized for structural elements, thin metal such as Al and Cu for electrical interconnects and passive layers such as SiN and SiO₂ for electrical insulation, and SiO₂ as sacrificial layers. Si is not only elastic and strong but is also a good piezoresistor and thermal conductor. Therefore, it is suitable for both mechanical and some transducer elements in MEMS. With similar mechanical properties to those of single crystalline Si, polycrystalline and amorphous Si are usually deposited as thin films and used as mechanical elements. The Si material set has been developed in parallel with microfabrication processes for microelectronics, ensuring that it allows integration with electronics as well as providing a high degree of process stability. Thus, Si-based MEMS still represent the overwhelming majority of commercial MEMS. However, the properties of the Si material set also restrict the applications that can be considered and there is great potential for advancing the field of MEMS by widening the materials available for their fabrication.

4.1. Materials for Micromechanical Components Micromechanical elements are key components in many MEMS devices such as suspended proof masses and springs in inertial sensors, diaphragms in pressure sensors, beam structures in resonators and stators/rotors in micromotors. In these systems, device performance is dictated by the mechanical properties of the structural materials, particularly; Young's modulus, fracture strength, residual stress and tribological properties. As shown in Table 2, microfabricated Si has a high strength-to-density ratio and a high strain to failure, making it a good candidate material. However, for devices required to operate at high temperatures, or for very high resonant frequency resonators materials with superior properties to Si are sought⁴⁹, with silicon carbide (SiC) and diamond as leading examples.

With the highest known specific stiffness of any material, diamond has attracted interest as an mechanical material for MEMS. Diamond resonators for radio frequency (RF) MEMS have demonstrated GHz resonant frequencies with ultrahigh quality factors^{50,51}. This opens up potential markets in the communications industries. However, in exploration of diamond as a MEMS material, one of the key challenges lies in the difficulty of integrating diamond films with other materials mainly due to its high deposition temperature. Also, the quality of most of the diamond films deposited still present rough surfaces and high internal stress due to the mismatch between the substrates and the diamond films, which are detrimental to MEMS devices. Recent findings suggest⁵² that ultrananocrystalline (UNCD) films have promising micromechanical, morphological and tribological properties, which could be better suited for MEMS devices than polycrystalline and amorphous diamond. A UNCD microturbine is shown in Fig. 7. Also, low temperature (550°C) deposition of UNCD has been developed⁵³ and efforts have been made to realise UNCD oscillators and resonators integrated

Figure 4: Materials Selection Chart (thermal expansion coefficient, α vs Young's modulus *E*) for bimaterial thermal actuators using a silicon substrate. Contours are for normalized moments (*M*), deflections (δ) and work (*W*).



with CMOS chips as a part of a joint collaboration between Advanced Diamond Technologies Inc. and Argonne National Laboratory⁵⁴. The main challenge towards future commercialisation of diamond MEMS is the development of cost-effective deposition techniques capable of large area and high volume production without compromising its material properties.

SiC is more advanced than diamond in terms of device development. It has also been recognised as an excellent candidate for microsensor and microactuator applications in harsh environments such as high temperature, high power level and strong corrosion, where Si is not suitable. Since Cree Research Inc. became the first supplier of SiC substrates in 1987, single crystal SiC wafers (4H and 6H) have been commercially available. In recent years, epitaxial grown single⁵⁵ and poly-crystalline⁵⁶ SiC layers on Si or SOI wafers have enabled large area SiC deposition which further stimulates the development of SiC MEMS prototypes with a view towards their commercialisation. In particular, SiC resonator structures have been realised using mechanical⁵⁷, electrostatic⁵⁸ and electrothermal⁵⁹ actuation, respectively. SiC can also be employed as the diaphragm⁶⁰ as well as piezoresistors⁶¹ in pressure sensor applications. Another important area of technology is SiC accelerometers, which are particularly attractive for detecting highg acceleration at elevated temperatures such as in aeroplane engines, military and space applications^{62,63}. From a commercialisation point of view, the field of SiC is currently in its infancy and occupies a niche MEMS market. However, companies such as FLX Micro offer a

limited commercial SiC foundry service. Also, for applications in harsh environments, it is particularly important to realise the integration between SiC MEMS and electronics. Thus, the development of SiC electronics has a direct impact on its MEMS counterpart.

One of the major reliability issues relating to Si MEMS is wear and unwanted adhesion. Due to the relatively low toughness of Si, fracture can occur when the structures are subject to high contact pressures. This can introduce debris and friction potentially leading to wear failure. Adhesion results from the dominance of surface forces due to the affinity of water or static from the structural components at the micrometer scale. Si micromechanical devices such as pinwheels⁶⁴, micromotors⁶⁵ and microturbines⁶⁶are often subject to stiction problems preventing start-up⁶⁷ and wear-related failure⁴⁹. Both diamond⁶⁸ and SiC⁶⁹ have been proposed as effective coating materials for Si micromechanical components in order to reduce friction and stiction and thereby achieve better reliability and enhancement of device lifetime.

4.2. Materials for Transducer Elements

Signal transduction from one physical domain to another is essential for the operation of MEMS sensors and actuators. Among them, electrical– mechanical and electrical–thermal–mechanical transductions are the most common transduction pathways and have been well demonstrated by the operation of electrostatic and electrothermal actuators and sensors. In electrical-mechanical transduction, external force can induce variation



Figure 5: Process Selection Chart for determining capabilities of various microfabrication processes to achieve a trench of depth, h, and width w.

in electrical signals due to displacement as in pressure sensors while input electrical energy can also generate mechanical function as in micro switches and motors. For a typical "U" shaped electrothermal actuator, input current passes through the structure and generates different ohmic heating between the two arms which leads to lateral force and then motion. Most materials such as Si and metals with a certain level of conductivity or with conductive components can be utilized as the structural elements in electrostatic or electrothermal applications. However, in many other cases, materials with specific properties are required to enable a particular transduction function. Many of these materials such as piezoelectric materials, thermal electric materials, magnetic materials and shape memory alloys (SMA) have enabled new avenues of applications in MEMS.

Crystalline materials that lack a centre of symmetry can exhibit piezoelectric properties. Thin film piezoelectric materials such as lead zirconate titanate (PZT)⁷⁰, zinc oxide (ZnO)⁷¹ and polyvinylidene fluoride (PVDF)⁷² are particularly attractive for MEMS applications because Si can be used as a substrate. Piezoelectric materials offer a large power density with low impedance and power requirements as well as an inherently high operating frequency, which are advantageous for MEMS actuators and sensors. A detailed review of the applications of piezoelectric materials in MEMS is provided in reference⁷³. Shape-memory

alloys undergo a temperature-induced phase change, resulting in a large volumetric strain, when heated above a critical transition temperature. Heating in MEMS can be readily achieved electrically. Shape memory alloys can exert larger forces than their piezoelectric and electrostatic counterparts which is particularly attractive for actuation functions. Among the broad range of SMA such as Ti/Ni, Cu/Al/Ni, Fe/Ni and Fe/Pt, thin film $Ti_x Ni_{1-x}$ alloys are the most widely used for MEMS actuators^{74,75} due to their simple composition and robustness. A TiNi micromirror structure is shown in Fig. 8. Thermal energy can also be directly converted into electrical energy and vice versa by thermoelectric materials. This material set can be particularly attractive for micro-cryogenic coolers based on the Peltier effect as well as power generators based on the Seebeck effect. Thin film thermoelectric materials are mainly considered for MEMS applications as they can be readily grown or deposited using common cleanroom facilities. In particular, the $(Bi_{1-x}Sb_x)_2Te_3$ compounds⁷⁶, Si_{1-x}Ge_x compounds⁷⁷ and polycrystalline Si^{78,79} have been explored. Inadequate energy conversion efficiency is still the main drawback related to the performance of these thermoelectric MEMS. Finally, in the presence of magnetic fields, magnetostrictive materials allow the interchange of mechanical and magnetic energies. The most advanced magnetostrictive materials such as $Tb_x Dy_{1-x} Fe_y$ (Terfenol-D) have been explored^{80,81} for MEMS



because they exhibit large magnetostriction at room temperature and require relatively small fields. However, the exploitation of magnetostrictive materials in MEMS is still very much restricted due to the relative low force and displacement output.

4.3. Polymer MEMS

Polymers especially synthetic polymers offer many advantages because they can be tailored to give a wide range of properties while requiring low temperature processing. A wide variety of polymer materials have been used for MEMS including; PMMA (polymethylmethacrylate), polyimide, photoresists, SU-8 resist, PDMS (polydimethylsiloxane), biodegradable polymers, parylenes, liquid crystal polymers and Teflon. One of the most important advantages of using polymer materials in MEMS applications is their low cost. It is relatively expensive to use Si wafers which have limited area as well as to create high aspect ratio (depth to width ratio) microstructures with conventional dry etching techniques. The requirement for low processing temperatures allows MEMS devices to be fabricated on large area or flexible substrates such as glass and plastics at high volumes. Thermoplastics and thermosets also allow low cost processing methods to be effectively used including; molding, embossing, melt processing, and imprinting. The inert properties of polymers such as polyimide and their relatively easy coating and bonding processes can also be advantageous for packaging purposes. The cost of polymer MEMS

can be reduced by a factor of ten over Si-based devices.

Meeting the biocompatibility requirements of biological and chemical applications, polymer materials such as polyimide, PMMA, SU-8 have been explored extensively for MEMS in the life sciences and medicine especially for microfluidic applications. Deep features can be economically fabricated in polymers. For instance, structures with heights of more than 1000 μ m can be formed in SU-8. The pre-patterned polymer layers can then be thermally bonded together to create sealed channels for microfuidic systems⁸², as shown in Fig. 9. Biocompatible parylenes have also been used in the form of coatings on implantable microelectrodes⁸³. Polymers with specific functionalities have also been utilized in MEMS sensors and actuators such as PVDF⁷² and piezoelectric polyimide⁸⁴. Electroactive polymers (EAP)⁸⁵, so called "artificial muscles", exhibit shape changes in response to electrical stimulation and thereby can be used as effective transducer components in MEMS. However, in order to raise the performance of these devices, a generic challenge is to develop polymers with a higher actuation stress capability, higher mechanical energy density and higher operating temperatures. For example, the current temperature range over which the piezoelectric properties of PVDF can be maintained is limited to be less than 80°C. Also, polymers are generally less stiff than conventional inorganic substrates such as Si and therefore they have limited applications for MEMS



Figure 7: An optical micrograph of an UNCD microturbine⁵².

in which micromechanical performance is critical. However, this also suggests that polymer MEMS require lower operating voltages in applications such as electrostatic actuators compared to their Si counterparts.

4.4. Nanostructured Materials in MEMS/NEMS

As described in Section 2, exceptional material properties often occur at extremely small scales, particularly in the nanometre range, because the feature size is of the same scale as the critical size for several physical phenomena. This has initiated the exploitation of nanostructured materials in NEMS devices. The defining characteristic of nanomaterials is that they have a feature size of 1–100 nm. Discrete nanomaterials such as CNTs and nanowires as well as continuous materials with nanomaterial compositions such as nanocomposites or ultrathin films can exhibit distinctive properties and thereby can be employed as building blocks in NEMS.

When discrete nanomaterials are used as structural materials, their extremely small size, large surface to volume ratio, specific functionalities or combinations of them can be utilized for NEMS to achieve advanced performance. Ultra high frequency micromechanical resonators have been reported using metal nanowires⁸⁶ and CNTs⁸⁷ to achieve resonant frequencies from hundreds of MHz to the GHz range. The tiny geometries and high mechanical strength herein play key roles. These resonators not only offer the potential for extreme mass and force sensitivity but also provide a possible approach to observe quantum phenomena directly. One-dimensional nanostructures such as Si^{88,89} or conducting polymer nanowires^{90,91} have also been used as biosensors. In this case, the electrical properties of nanowires are strongly influenced by minor perturbations because of their high surface-to-volume ratio and tunable electron transport properties due to the quantum confinement effect. Recent studies⁹² of piezoelectric nanowires such as ZnO have also demonstrated the potential to harvest vibration energy via an electrical generator, which could be utilized in future selfpowering nanodevices. However, when discrete nanomaterials are used in NEMS, it is important to anticipate significant and generic challenges in sample handling and device testing.

Continuous nanostructured materials can be formed by the incorporation of nanomaterials into conventional material matrices or combinations of different nanomaterials. The overall material properties can then be improved and tailored over a wide range by tuning the relative densities of the compositions and thereby generating desirable materials suitable for certain applications. For instance, theoretical studies suggest⁹³ nanofibre (nanotubes, nanorods and nanowires) reinforced composites could exhibit relatively low modulus and high wave speed. The combination of these conflicting properties is required by some MEMS applications where low actuation force or actuation voltage and high actuation frequency are required⁹⁴. Ceramic nanocomposites have also been proven⁹⁵ to be more effective than their conventional ceramic counterparts in MEMS applications due to the increased fracture toughness. As another important continuous nanomaterial set, ultrathin films with submicrometer thickness have been utilized as structural materials for NEMS sensors in order to achieve ultrahigh sensitivity and fast response⁹⁶. However, these systems usually are susceptible



Figure 8: TiNi micromirror structure with a Si cap acts as top mirror and the arms fabricated with TiNi/Si beam structure⁷⁵.

to mechanical energy loss due to the increase in surface to volume ratio and thereby appropriate surface treatment and a passivation process needs to be applied⁹⁷. Nanocomposite ultrathin films have also been investigated for NEMS applications. A recent result⁹⁸ demonstrated the possibility to engineer metal nanocomposites to achieve nearly atomically smooth surfaces, high stiffness and high electrical conductivity. These metal nanocomposite ultrathin films shows advantageous for applications such as NEMS switches. Generically, continuous nanostructured materials usually require optimisation processes to obtain materials with the desired properties as well as developing effective micromachining techniques before they can be exploited in MEMS.

5. MEMS Fabrication Processes

Originally developed in the microelectronics industry, microfabrication processes are essential to the creation of functional MEMS devices and are often the major constraint to realising commercial MEMS products. Furthermore, the development of the MEMS material set largely depends on the availability of processing techniques. Thus, developments of effective fabrication processes are not only important for conventional Si MEMS but are also critical for the realisation of MEMS with new materials.

Based on micromachining dimensions, fabrication processes are often coarsely categorised into bulk and surface machining and detailed in most of MEMS textbooks¹⁵. Bulk micromachining refers to process sequences in which threedimensional features are created in a substrate for the purpose of diaphragm or cavity formation. On the contrary, surface micromachining mainly involves deposition and etching of relatively thin layer materials on bulk substrates. Throughout their evolution MEMS material and fabrication processes have evolved from well-established microelectronic processes. A recent example is the increasing usage of silicon on insulator (SOI) substrates in MEMS devices stimulated by the fact that SOI has entered into the mainstream of the microelectronic industry^{99,100}. However, the continuous increase of Si substrate dimensions in the microelectronic industry has started to bring negative consequences for the MEMS industry. Tool sets to accommodate large Si wafers will inevitably be developed and dominate the mainstream microfabrication processes. But most MEMS currently are fabricated using relatively small substrates (100 mm or 150 mm in diameters) and many of these are still in prototype stages. To address this issue, it is important to establish mature MEMS fabrication processes particularly IC-compatible processes. Furthermore, the continuous emergence of novel MEMS material sets for advanced device applications has also posed new challenges towards microfabrication techniques. This section focuses on the key fabrication processes and their associated issues in MEMS.

5.1. Lithography for MEMS

Lithography is the first step for patterning. As described in most microfabrication text books¹⁵, optical lithography using e.g. deep-ultraviolet, x-ray, excimer laser as energy sources has been the enabling technologies for virtually all integrated circuits (ICs) and MEMS productions to-date. For decades, the continuous shrinkage of electronic

Figure 9: Cross section of a microchannel made of SU-8 [82].



circuits has pushed optical lithography to achieve increasingly small line widths. In the semiconductor industry, currently 90 nm features are already in mass production of electronic devices and 29.9 nm high quality line patterns have been recently reported¹⁰¹. These developments also provide effective high resolution lithography techniques for MEMS. However, unlike IC devices which only require two dimensional or planar structures to be fabricated, MEMS usually contain 3D features. This has posed new challenges to develop novel lithography techniques to suit special MEMS patterning requirements. As a result, "soft lithography" which encompasses imprinting techniques for polymer MEMS has been developed and proven to be cost-effective and advantageous to create features with high aspect ratios. Also, technologies such as holographic lithography^{102,103} stereo lithography¹⁰⁴ and gray-scale lithography¹⁰⁵ are particularly attractive for the formation of 3D MEMS structures. Techniques using numerically controlled e-beam writing¹⁰⁶, flexible stamp¹⁰⁷, flexible mask¹⁰⁸, shadow-masking¹⁰⁹ and laminated film resist¹¹⁰ have been employed to produce microstructures on nonplanar surfaces and even deep trenches. With optical lithography reaching its practical limit, the lithography techniques originally derived from MEMS devices such as nanoimprint could transfer to future generation microelectronic products.

5.2. Pattern transfer techniques

Following pattern definition by lithography, these patterns are transferred into structural materials to create microfeatures. The pattern transfer processes usually involves removal or addition of materials or both, which are described in the following subsections, respectively.

5.2.1. Etching processes

The most prevalent material subtractive method for pattern transfer in MEMS is based on etching processes which are usually categorized as wet and dry etching. Wet etching utilizes suitable liquid chemicals to attack and remove the exposed substrate regions while dry etching usually takes place in chemically reactive vapour or reactive species in glow-discharge plasma. Both wet and dry etching are extensively used in Si bulk and surface micromachining and their standard processes are well-described elsewhere¹⁵. An important recent dry etching innovation is the emergence of XeF₂ which etches Si spontaneously with an isotropic etch profile as shown in Fig. 10^{111} . Using XeF₂ in its gaseous form (non-plasma), high Si etch rates (up to 15um/min) have been achieved with extreme etching selectivity over Al, SiO₂, SiN and even photoresist¹¹¹. This implies that existing CMOS electronics can be protected during the etch process and therefore suggests it is potentially suitable for post-CMOS etching. For plasma related dry etching, the Bosch process¹¹² is still extensively used to achieve Si high aspect ratio features in which repetitive etch and passivation steps are utilized during the cyclic process. Due to the involvement of both physical and chemical reactions in the plasma, plasma etch processes can often be tuned into either the physical- or chemical-dominated regimes which can be particularly attractive in the developments of effective etching processes for new MEMS material sets such as SiC^{113,114}.

5.2.2. Additive processes

Common additive processes in MEMS mainly involve material growth and/or thin film deposition. MEMS structural layers such as polysilicon, metal electrodes/interconnects and sacrificial/insulation layers such as SiO₂ and SiN can be deposited by chemical vapour deposition (CVD), physical vapour deposition (PVD), or electrodeposition (or electroplating). These additive processes have also been extensively explored towards the development of new material sets that can be utilized in MEMS applications. It is worth noting that additive processes associated with high temperatures hinder the development of cost-effective MEMS in terms of integration with ICs. For example, polycrystalline Si is still commonly used as a MEMS structural material. However, the high temperature (over 600°C) required for its CVD deposition is incompatible with CMOS electronics. Therefore, for future MEMS and IC integration, it is important to further develop low temperature additive processes or CMOS compatible materials. Micromolding is another important additive process



Figure 10: SEM images of XeF₂ etched pit openings to illustrate its isotropic etch profile¹¹¹.

in which materials are deposited into pre-patterned micromolds followed by removal of the mold along with unwanted structural materials. LIGA was the first MEMS micromolding process. In LIGA metal is electroplated into a polymer mold preformed by high energy x-ray lithography. It is an effective technique by which high aspect ratio structures with good side-wall controls can be fabricated¹¹⁵⁻¹¹⁷. Micromolding allows the formation of MEMS components from a range of materials with flexible geometry especially those that are difficult to etch. Thus, it is particularly attractive for MEMS prototyping. For instance, using prepatterned polysilicon mold, SiC micromotors¹¹⁸ have been fabricated, as shown in Fig. 11. Despite these advantages, micromolding is still not at the level of high volume production and commercialisation. This is partially due to insufficient knowledge in process control, difficulty in reproducibility¹¹⁹ and also requirement to use highly collimated synchrotron sources in the case of LIGA¹²⁰.

5.3. Wafer bonding processes

Wafer bonding is an important technique for material and microsystem integration. It enables the formation of SOI substrates¹²¹, sealed microstructures/cavities¹²² and the integration of devices at different levels^{123,124} and dissimilar materials^{125,126}. Therefore, it has been identified as a promising process for the realisation of future three-dimensional integrated circuits (3D IC) in microelectronics as well as for the creation of multilayered systems, 3D microcavities/microchannels and effective packaging in MEMS. Wafer bonding techniques utilized in MEMS can be classified into direct bonding (fusion bonding), anodic bonding and intermediate-layer bonding. As depicted in many articles¹²⁷ and books^{128,129}, the direct bonding process is used to mechanically join two Si wafers together by creating hydrophobic or hydrophilic surfaces that are brought into contact and annealed at high temperatures. Anodic bonding joins a Si wafer to an alkali glass (e.g. pyrex borosilicate glass) wafer by the aid of charge migration driven by an applied electrostatic field. Intermediate-layer bonding includes eutectic, glass-frit and thermal compression bonding which utilized intermediate layers such as metal, glass or polymers to adhere the substrates together.

In general, wafer-bonding utilises a three step sequence: surface preparation, aligned bonding and annealing. Wafer bonding quality is strongly dictated by wafer surface conditions in terms of contamination, roughness and flatness. For instance, generally Si wafers with very few particles, ≤ 5 Å roughness and $\leq 5 \,\mu$ m flatness on 100 mm wafers are necessary to ensure a good direct bond¹³⁰. Combined with extensive post-cleaning processes, chemical-mechanical polishing (CMP)¹³¹ has been a common technique to create sufficiently flat and smooth surfaces to achieve high reliability bonds. Furthermore, wafer bonding alignment is commonly achieved by a variety of optical means, however alignment accuracy of a micron at best is currently obtained due to the large

optical structural loop¹³². These optical means are fundamentally limited by the wavelength of light and practically limited by the mechanical positioning systems required to match the two wafers. However, nanoprecision bonding alignments is necessary to ensure desired functions of multilayered MEMS¹²⁴ and electronic devices¹³³. Based on kinematic and elastic averaging effects, an innovative micromechanical method¹³⁴ has been recently proposed and the proof-of-concept results at the Si chip level demonstrated better than 200 nm bonding alignment accuracy without using any optical alignment facilities. Fig. 12 shows schematic drawings of the utilized micromechanical alignment features at the chip level. Finally, direct Si or SiO₂ bonding requires high temperature annealing $(\sim 800^{\circ}\text{C})$ to attain a strong and permanent bonding. The use of high temperature is not always acceptable, especially when there are low temperature materials (e.g. Al, polymers) or pre-fabrication electronic devices involved in the systems. Consequently, low temperature bonding techniques (lower than 400°C) have been widely developed. In most cases, plasma treatments¹³⁵ particularly oxygen plasmas¹³⁶, are applied to activate the surfaces before bonding which lead to a high bonding energy after annealing at low temperatures.

5.4. Packaging Issues

MEMS devices usually have to be packaged to provide electrical contact, mechanical protection as well as interfaces with the environment for sensing, interconnection and actuation. Hermetic seals are often required in MEMS packaging for environmentally sensitive components or fragile moving elements to ensure appropriate protection, assembly and long term reliable operations. In general, materials such as ceramics, metals and polymers are used in MEMS packaging as detailed elsewhere¹³⁷. However, unlike well-established IC packaging, MEMS packaging still faces significant technology barriers and cost issues. This has been a major obstacle to the commercialization of many MEMS products. The difficulty in MEMS die handling, the high level of protection required and custom packaging for each application have led to high package cost which may be as high as 80% of the product cost¹³⁸. In recent years, wafer level packaging (WLP) has become a promising alternative for high volume and low cost MEMS production^{137, 139} and is widely accepted by MEMS manufactures. WLP utilizes wafer bonding techniques to encapsulate devices. It allows MEMS devices and packaging to be manufactured and tested on wafers prior to singulation. The economic advantage of WLP will be more prominent with the increase of wafer size and shrinkage of MEMS die dimensions. Furthermore, since MEMS packages can also consume (absorb) and emit materials (outgas), to help control/maintain the atmosphere/vacuum within hermetic packages, the integration of effective getter materials into MEMS packages has also emerged as a promising technology 90, 140.

6. Materials Characterization

The success of the microelectronics industry has been underpinned by the reliability of the simulation tools available and the extremely well characterized electronic properties of the materials being utilized and the processes with which the products are created. For MEMS to achieve their promise of

Figure 11: SiC wobble micromotor formed using a sacrificial polysilicon mould¹¹⁸.







low unit cost and large volume production it is important that similar confidence exists in the output of design codes, which in turn implies that the capabilities to characterize the material properties are well proven. Given that electronic and other functional (e.g. optical and magnetic) properties receive considerable focus for mainstream applications (microelectronics, optoelectronics and memory devices), the most important MEMSspecific requirements for characterization are of the mechanical properties.

The key issue is that microfabricated materials have mechanical properties that are highly dependent on the fabrication route used to create them and the scale of the structures in which they are employed. As noted in section 2 properties at the microscale can vary considerably from those measured on bulk samples of material at the macroscale. In order to fully realize the potential for accurate and rapid simulation tools for design of MEMS models are required which link the fabrication route used to the value of the material property achieved via the microstructure and composition of the material. The first step towards this is to develop standard test methods with which to characterize the mechanical properties of microfabricated structures produced by the same processes and at the same scales as the intended application. This enables the creation of validated material property and process data-bases and correlations to permit simulation-based design. The following sections illustrate where progress in this direction has been made. There have also been several recent reviews of this area^{141,142}.

6.1. Elastic Properties

Measurement of elastic properties at small scales is relatively mature and can generally be achieved accurately. Cantilever beams, double

clamped-beams143 and diaphragms which are loaded electrostically¹⁴⁴, mechanically by external means (such as nanoindentor tips)¹⁴⁵ or by fluid pressure¹⁴⁶, with deflections measured by means of capacitance or optical sensors have been notably popular for this purpose. Raman spectroscopy can also be used to obtain strain distributions in silicon specimens¹⁴⁷. Tensile tests have been developed with interferometric strain measurement methods¹⁴⁸. Resonant structures have also been utilized¹⁴⁹ and offer the potential for particularly accurate measurements. These methods have allowed reproducible evaluation of the Young's moduli of deposited thin film materials. In addition the development of focused ion beam machining techniques have permitted test specimens to be fabricated without recourse to wafer-level patterning. The use of the unloading compliance during the nanoindentation of unpatterned thin films has received considerable interest as a method for extracting elastic properties, however this is not as accurate for this purpose as the methods described above. It does however have advantages in the extreme simplicity of the measurement. Hitherto relatively little work has focused on obtaining other elastic constants such as Poisson's ratios¹⁵⁰ and shear moduli or the thermal expansion coefficients¹⁵¹ and on understanding the possible effects of anisotropy due to the crystallographic texture of deposited thin films. It is noteworthy that a decade ago, even for a widely used material, such as polysilicon, values of moduli ranging from 132 GPa to 174 GPa were reported in the literature¹⁵² on material deposited by nominally identical processes. Test techniques, metrology and understanding of likely sources of error have improved significantly in the interim to the point where such measurements can be considered to be routine¹⁵³. Another significant recent development

Figure 13: A MEMS Fatigue test structure (Courtesy S. Brown).



is the introduction of test techniques that permit accurate property measurement without recourse to highly sophisticated instrumentation¹⁵⁴. This is a key step to migrate mechanical characterization of materials for microsystems into routine industrial practice.

6.2. Strength Characterization

The characterization of the strength of microfabricated materials and structures is required for microsystems that are designed to operate at high mechanical power densities and/or large deflection levels. The performance of such devices is limited by the strength of the materials of construction. Since the strength of materials, can be very dependent on the scale and the fabrication route, it is critical that measurements to be used for design purposes are obtained from test structures fabricated by the same processing route and at a similar scale to that to be used for the application for which they are intended.

Various approaches have been taken to obtain room temperature strength related properties. Nanoindendation has proven to be a viable means to extract information regarding plastic constitutive behavior¹⁵⁵ although it struggles in cases where there is anisotropy or heterogeneity. Electrostatic actuation has been used to generate forces sufficient to cause fracture in surface micromachined structures¹⁵⁶. However, in order to generate sufficiently high stresses to cause fracture by such means the cross section of the part typically has to be limited to a small fraction of the area used to generate the electrostatic force, and even then significant stress concentrations need to be introduced. In order to test larger specimens at higher force levels various workers have used mechanical loading applied via modified microhardness indentors¹⁵⁷ or nano-indentors to generate bending stresses to cause failure¹⁵⁸. In addition tensile tests have been performed using mechanical or electrostatic gripping and in situ strain measurement¹⁴⁸. These approaches are particularly necessary for the thicker structures realized by bulk micromachining and/or SOI layers. In the case of brittle materials, the statistical variation of strength is a key variable. Although Weibull statistics has been applied¹⁵⁹, it does not seem to adequately represent the scaling of strength into structural designs. This is most likely due to the strong interactions between the specimen geometry, the processing route and the resulting flaw population. Further work is required on this topic.

Obtaining elevated temperature properties for microfabricated materials is important as the MEMS devices are designed for high temperature applications, as well as to help develop models for microfabrication processes which utilize elevated temperatures for bonding or annealing. Bulge tests of pressurized cavities¹⁶⁰ have been used as one means of obtaining such data, as well as more conventional macroscale bend tests¹⁶¹ and compression tests¹⁶².

6.3. Adhesion and Bond Strength

Virtually all MEMS consist of multiple layers of materials created by deposition or bonding operations. The structural integrity of the bonds between layers is a key parameter in determining reliability. Several techniques are well established for measuring thin film adhesion including bulge testing¹⁶³, peel testing and residual stress driven cohesion measurements¹⁶⁴ and these are not unique to MEMS devices, although it is worth noting that microfabrication techniques play a key role in creating the test structures which allow these measurements. Of direct relevance to MEMS are test techniques used to measure inadvertent adhesion due to "stiction". The most well developed techniques use arrays of cantilever beams of defined stiffness, and then observe the critical length at which they adhere to the surface¹⁶⁵. Such methods are important in order to allow the development of surface modification and process sequences that eliminate the danger of stiction, which is a key problem for compliant surface micromachined structures.

As previously noted wafer bonding is of more specialized application to MEMS. A number

of techniques have been developed to allow determination of bond quality and strength. Nondestructive methods, including infra red, ultrasonic and X-ray imaging have been employed to detect macroscopic voids¹⁶⁶. This is particularly valuable during the initial contacting phase of fusion bonding operations since poor bonds can be indentified and the wafers separated and rebonded before the elevated temperature annealing step is carried out. Bond strength has been characterized by a number of techniques, including pressure burst testing, double cantilever beam specimens¹⁶⁷ and other mechanically loaded structures, which expose the bond to combinations of tension and shear stresses. More recently attention has focused on developing the ability to map bond toughness across a bonded pair of wafers using smaller micromachined chevron-notch specimens, which can be tested individually¹⁶⁸. Other work has also used patterning to measure the bonding energy (as opposed to the fracture energy)¹⁶⁹. Given the importance of bonding operations to MEMS fabrication this is a fertile area for materials science and mechanical engineering advancement.

6.4. Residual Stresses

Since MEMS devices typically contain several deposited and bonded layers of dissimilar materials residual stresses can play an important role in determining the reliability of the processes and the fabricated devices. The issues of thin film residual stresses have received considerable attention due to their importance in the microelectronics industry, and to a large degree these issues are the same as those found in MEMS. However, as MEMS devices are created which have larger mechanical power and force capabilities, thicker deposited layers are being investigated than are typically utilized in microelectronic applications. This is particularly true in devices which use molding operations, such as LIGA and CVD deposition of SiC. These thicker layers have a greater tendency to fracture and the thickness (and therefore size of the device that can be realized) may be limited by the residual stress state. The ability to control and characterize residual stresses is very important for the development of higher performance MEMS and various novel test structures have been created to permit residual stress characterization^{170,171}.

6.5. Fatigue

MEMS devices may be subject to very high numbers of fatigue cycles during their service lifetimes due to their inherently high operating frequencies. This raises the possibility of fatigue being a limiting factor on the allowable stress levels or useful life. These concerns have resulted in the recent development of test structures to probe the fatigue behavior of microfabricated materials. Typically these structures utilize electrostatic loading and excitation at resonance to obtain stress levels sufficient to cause fatigue failure. Such a structure is shown in figure 13. Such test methods have shown the possibility of fatigue processes in both ductile^{172,173} and brittle microfabricated materials¹⁷⁴. It is a matter of considerable discussion as to whether the mechanism observed in brittle materials, particularly polycrystalline silicon is a cyclic fatigue process¹⁷⁵, or rather an environmentally assisted slow crack growth process, albeit with some synergy between load cycling and crack growth¹⁷⁶. It is also worth noting that many commercial accelerometers and pressure sensors have experienced extremely high numbers (> 10^{10}) of cycles apparently without sustaining any fatigue failures, and at least one major company has concluded that fatigue is not of sufficient concern for their polysilicon sensors to merit direct consideration in their design process. However, as MEMS devices start to push towards higher mechanical power levels fatigue may increasingly become a concern.

6.6. Surface Forces and Tribology

The high surface area to volume ratio of MEMS devices implies that tribological effects are likely to be important factors in determining performance. Experiences with surface micromachined accelerometers¹⁷⁷ and micromotors¹⁷⁸ suggest that surface adhesion due to charge build up or moisture adsorbtion is a critical issue that results in stiction and hysteresis. The same scaling of electrostatic forces that makes it attractive for prime movers at the microscale also can prove a liability. In addition the use of a wet etch as the release step can be complicated by the introduction of capillary forces between elements that prevent their separation. Experience with micromotors and micro-gear trains running at high rotational speeds on unlubricated sliding contacts has indicated that wear processes are very important in both allowing the bearing surfaces to be worn in to allow low friction operation, and subsequently in contributing to failure. This is despite the very low inertial and gravitational forces associated with the devices.

The importance of tribology for MEMS has resulted in a growing literature on the subject¹⁷⁹, quantitative measurements of surface adhesion forces, friction and wear, and erosion behavior have been obtained from a variety of devices. Attempts are being made to modify micromachined surfaces¹⁸⁰ or apply low friction coatings¹⁸¹ in order to promote better tribological characteristics and there is a great need for increased understanding in this area if reliable and durable devices are to be created. In addition non-materials solutions, involving the use of air bearings¹⁸² or magnetic levitation offer promise for overcoming some of the tribological issues associated with high speed MEMS.

The mechanical characterization of materials is a key activity if reliable and durable MEMS devices are to be realized. Addressing the topic of durability is particularly important if MEMS are to be used in embedded systems for system health monitoring and other safety critical applications. It will also become extremely important to be able to perform failure analyses in cases where structural integrity is lost. This is not a trivial activity at the microscale.

7. Concluding Remarks

Since 2000 the research activity in materials issues associated with MEMS has expanded dramatically. The number of research articles associated with the topic has increased by more than an order of magnitude and the annual rate of publication of research articles has increased by a commensurate factor over the past decade. The manifestation of this activity can be seen in all the areas considered in the present article. The MEMS material set has significantly expanded, and is no longer limited by the constraints of CMOS-compatible materials and fabrication facilities. Similarly the range of processes available has increased beyond those that are solely the province of conventional CMOS. The characterization techniques for mechanical properties have matured and several are being used, or considered for use, in industry. Furthermore microfabrication and microfabricated structures are routinely used to characterize materials, particularly thin films for microelectronic applications and biological materials. Notwithstanding the significant progress in the development of new materials and techniques associated with MEMS it is noteworthy that the vast majority of commercial MEMS still are produced with the CMOS-compatible material and process set. This reflects the massive investment of financial capital in fabrication facilities and intellectual capital in the scaling up of massproduction processes. It is also significant that no-truly nanomechanical systems have emerged, which partially reflects unfavourable scaling for many mechanical performance metrics below the 1 μ m scale, although these constraints do not apply for non-mechanical integrated systems. It is clear however that the foundations have now well and truly been laid for the irreversible broadening of the MEMS materials and process set. It is also significant that some processes which were

primarily developed for MEMS devices, such as multi-wafer bonding and deep anisotropic etching are being increasingly considered for mainstream microelectronic devices, particularly with regard to 3-D interconnect schemes. There is no sign of the rate of decline in the rate of progress in materials and processes for MEMS, which suggests that the coming seven years will witness at least as much technical progress as the previous seven.

Received 31 July 2007; revised 14 August 2007.

References

- 1. S. D. Senturia, MicroSystem Design, Kluwer Academic Publishers, Boston, 2001.
- G.T.A. Kovacs, Micromachined Transducers Sourcebook, WCB/McGraw-Hill., Boston, 1998.
- J. J. Allen, Micro electro mechanical system design, Taylor & Francis/CRC Press, Boca Raton 2005.
- A. H. Epstein, S. D. Senturia, Microengineering Macro power from micro machinery, *Science* 276, 1211, 1997.
- K. F. Jensen, "Silicon-based microreactors", Microreactor Technology and Process Intensification ACS Symposium Series 914, 2–22 2005.
- D.L. Polla, A.G. Erdman, W.P. Robbins, D.T. Markus, J.Diaz-Diaz, R. Rizq, Y. Nam, H. T. Brickner, A. Wang, P. Krulevitch, "Microdevices in medicine" *Annual Review of Biomedical Engineering*, 2551–576, 2000.
- A. Khademhosseini, R. Langer, J. Borenstein, J. P. Vacanti, "Microscale technologies for tissue engineering and biology" *Proc. Natl. Acad. Science. USA.* 103, 2480–2487, 2006.
- B. W. Cook, S. Lanzisera, K. S. J. Pister, SoC issues for RF smart dust, *Proc. IEEE* 94 (6): 1177–1196, 2006.
- A. A. Berlin, K. J. Gabriel, Distributed MEMS: New challenges for computation, *IEEE Computational Science and Engineering*, 4, 12–16, 1997.
- Y. Fukuta, Y-A. Chapuis, Y. Mita Y., H. Fujita H. Design, fabrication, and control of MEMS-based actuator arrays for air-flow distributed micromanipulation, *J. MEMS*, 15, 912–926, 2006.
- S.M. Spearing, "Materials issues in microelectromechanical systems (MEMS)", Acta Mater. 48, 179–196, 2000.
- 12. Integrated Sensors, Microactuators, and Microsystems (MEMS), Ed. K. D. Wise, *Proc. IEEE*, **86**, 1998.
- T. Fukuda , W. Menz, Handbook of sensors and actuators ; v. 6 Micro mechanical systems : principles and technology, Elsevier, Amsterdam, New York, 1998.
- 14. *Micromechanics and MEMS : classic and seminal papers to 1990*, Ed. Trimmer, W., IEEE Press, New York, 1997.
- M. J. Madou, "Fundamentals of microfabrication", CRC press, Boca Raton, FL, 2002.
- C. T. –C. Nguyen, "Vibrating RF MEMS for next generation wireless applications", *IEEE Proc. Custom integrated circuits* conference, pp. 257–264. 2004.
- B. Ilic, Y. Yang, K. Aubin, R. Reichenbach, S. Krylov, and H. G. Craighead, "Enumeration of DNA Molecules Bound to a Nanomechanical Oscillator", *Nano Letters*, 5, 925–929, 2005.
- H. G. Craighead, "Nanoelectromechanical Systems", Science, 290, 1532–1535, 2000.
- Xinxin Li, Takahito Ono, Yuelin Wang, Masayoshi Esashi, "Study of ultra-thin NEMS cantilevers-high yield fabrication and size-effect on Young's modulus of silicon" *IEEE Proc. of the* 15th *IEEE international conf. Micro Electro Mechanical Systems*, 427–430, 2002.
- 20. B. Ilic, D. Czaplewski, H. G. Craighead, P. Neuzil, C. Campagnolo and C. Batt, "Mechanical Resonant

Immunospecific Biological Detector", *Appl. Phys. Lett.*, 77, 450–452, 2000.

- Liudi Jiang, R. Cheung, M. Hassan A. J. Harris, J. S. Burdess, C. A. Zorman, M. Mehregany, "Fabrication of SiC microelectro-mechanical systems (MEMS) using one-step dry etching", J. of Vac. Sci. and Tech. B 21, 2998, 2003.
- H. B. Peng, C.W. Chang, S. Aloni, T. D. Yuzvinsky, and A. Zettl, "Ultrahigh Frequency Nanotube Resonators", *Physical Review Letters*, Vol. 97, pp. 0872–03, 2006.
- N. Maluf, K. Williams, "An Introduction to Microelectromechanical System Engineering," pp. 96, Artech House Inc., 2004.
- E. Arzt. "Size effects in materials due to microstructural and dimensional constraints: a comparative review", Acta Mater., 46, pp. 5611, 1998.
- H. A. Stone and S. Kim, "Microfluidics: Basic Issues, Applications, and Challenges," AIChE Journal, Volume 47, Issue 6, pp 1250–1254. 2001.
- R. P. Leland, "Mechanical-Thermal Noise in MEMS Gyroscopes", IEEE Sensors Journal, 5, 493–500, 2005.
- Mou Jianqiang , Chen Shixin , Lu Yi , Lim Boon Buan, "Thermal Stability Evaluation of MEMS Microactuator for Hard Disk Drives", Digest of the Asia-Pacific Magnetic recording Conf., pp. WE-P-13-01 - WE-P-13-02, 2002.
- G. Li and A. A. Tseng , "Low Stress Packaging of a Micromachined Accelerometer", *IEEE Trans. Electronics Packaging Manufacturing*, 24, 18–25, 2001.
- 29. A. C. Imhoff, "Packaging technologies for RFICs: current status and future trends," *Proc. IEEE Radio Frequency Integrated Circuits (RFIC) Symposium*, 7–10, 1999.
- H. Reichl, V. Grosser, "Overview and development trends in the field of MEMS packaging," *Proc. The 14th IEEE International Conf. on MEMS*, pp. 1–5, 2001.
- H. de Lambilly, R. Grace, and K. Sidhu, "Package or Perish", Proceedings of the 1996 Sensors Expocon, p. 275, 1996.
- J. Wang and J-S. Wang, "Carbon nanotube thermal transport: Ballistic to diffusive", *Appl. Phys. Lett.* 88, 111909, 2006.
- E. Brown, L. Hao, J. C. Gallop and J. C. Macfarlane, "Ballistic thermal and electrical conductance measurements on individual multiwall carbon nanotubes", *Appl. Phys. Lett.*, 87, 023107, 2005.
- M. M. Treacy, T. W. Ebbesen, J. M. Gibson, "Exceptionally high Young's modulus observed for individual carbon nanotubes", *Nature* 381, 678–680, 1996.
- H. B. Peng, C.W. Chang, S. Aloni, T. D. Yuzvinsky, and A. Zettl, "Ultrahigh Frequency Nanotube Resonators", Phys. Rev. Letters, Vol. 97, pp. 087203, 2006.
- Kong J, N. R. Franklin, C. Zhou, M. G. Chapline, S. Peng, K. Cho and H. Dai, *Science* 287, 622, 2000.
- Inpil Kang, M. J Schulz, J.H. Kim, V. Shanov and D. Shi, "A carbon nanotube strain sensor for structural health monitoring", *Smart Mater. Struct.* 15, 737–748, 2006.
- J. M. Kinaret, T. Nord, and S. Viefers, "A carbon-nanotubebased nanorelay", *Appl. Phys. Lett*, 82, 1287–1289, 2003.
- X. M. H. Huang, C.A. Zorman, M. Mehregany, M.L. Roukes, *Nature*, 421, 496, 2003.
- M. F. Ashby, "Materials Selection in Mechanical Design" Butterworth-Heinemann, Oxford UK, 1999.
- V. T. Srikar and S. M. Spearing, "Materials selection for microfabricated electrostatic actuators" *Sensors and Actuators*, *Part A, Physical*, **102**, 279–285, 2003.
- V.T. Srikar and S. M. Spearing, "Materials selection in Micromechanical Design: an application of the Ashby Approach" J. MEMS, 12(1) 3–10, 2003.
- M. F. Ashby, "Multi-Objective Optimization in Materials Design and Selection" Acta Mater. 48 359–369, 2000
- 44. S. Prasanna and S. M. Spearing, Materials selection and design of microelectrothermal bimaterial actuators, *J. MEMS*

16, 248–259, 2007.

- S. Prasanna, S. M. Spearing, "Effect of Heat Transfer on Materials Selection for Bimaterial Electrothermal Actuators", *JMEMS*, in press
- S. Prasanna, S. M. Spearing, "Optimal Materials Selection for Bimaterial Piezo Micro Actuators" *JMEMS*, Submitted
- D. J. Quinn, S. M. Spearing, M. F. Ashby and N. A. Fleck "A Systematic Approach to Process Selection in MEMS", *J. MEMS*, 15 (5): 1039–1050, 2006.
- D. J. Bell, T. J. Lu, N. A. Fleck and S. M. Spearing "The Selection of MEMS Actuators and Sensors", *J. Micromech. Microeng.* 15, S153–S164, 2005
- K. J. Gabriel, F. Behi, R. Mahadevan and M. Mehregany, Sensors Actuators A 21–23 184, 1990.
- L. Sekaric, J. M. Parpia, H. G. Craighead, T. Feygelson, B. H. Houston, J. E. Butler, "Nanomechanical Resonant Structures in Nanocrystalline Diamond," *Applied Physics Letters*, 81, 4455, 2002.
- J. Wang, J. E. Butler, T. Feygelson, T.-C. C. Nguyen, "1.51-GHz Nanocrystalline Diamond Micromechanical Disk Resonator with Material-mismatched Isolating Support," *Proceedings*, *17th Int. IEEE Micro Electro Mechanical Systems Conf.*, Maastricht, The Netherlands, Jan. 25–29, 641–648, 2004.
- 52. O. Auciello, J. Birrell, J. A Carlisle, J. E Gerbi, X. Xiao, Bei Peng and H. D. Espinosa, "Materials science and fabrication processes for a new MEMS technology based on ultrananocrystalline diamond thin films", *J. Phys. Condens. Matter* 16 R539–R552, 2004.
- 53. S. P. Pacheco, P. Zurcher, S. R. Young , D. Weston, W. J. Dauksher, O. Auciello, J. A. Carlisle, N. Kane, and J. P. Birrell, "Characterization of low-temperature ultrananocrystalline diamond RF MEMS resonators" *IEEE proc. Gallium Arsenide and other semiconductor application symposium*, 2005.
- 54. "Diamond Technology to Revolutionise Mobile Communications" http://www.nanotechnologyworld.co. uk/content/view/298/
- Zorman, C. A., Fleischman, A. J., Dewa, A. S., Mehregany, M., Jacob, C., Nishino, S. and Pirouz, P. Epitaxial growth of 3C– SiC films on 4 in. diam (100) silicon wafers by atmospheric pressure chemical vapor deposition' *J. Appl. Phys.*, 78, 5136, 1995.
- Zorman, C. A., Rajgopal, S., Fu, X. A., Jezeski, R., Melzak, J. and Mehregany, M. 'Deposition of polycrystalline 3C-SiC films on 100 mm diameter (100) Si wafers in a large-volume LPCVD furnace' *Electrochem. Solid State Lett.*, 5, 99, 2002.
- Liudi Jiang, R. Cheung, M. Hassan, A. J. Harris and J. S. Burdess, C. A. Zorman and M. Mehregany, "Dry release fabrication and testing of SiC electrostatic cantilever actuators" *Microelectronic Engineering*, 78–79, 106–111, 2005.
- Liudi Jiang, R. Cheung, M. Hassan A. J. Harris, J. S. Burdess, C. A. Zorman, M. Mehregany, "Fabrication of SiC microelectro-mechanical systems (MEMS) using one-step dry etching", *Journal of Vac. Sci. and Tech.* B 21, 2998–3005, 2003.
- Liudi Jiang, R. Cheung, J. Hedley, M. Hassan, A. J. Harris, J. S. Burdess, M. Mehregany and C. A. Zorman, "SiC cantilever resonators with electrothermal actuation", *Sensors* and Actuators A 128, 376–382, 2006.
- D. J. Young, J. G. Du, C. A. Zorman, and W. H. Ko, "High-Temperature Single-Crystal 3C-SiC Capacitive Pressure Sensor" *IEEE Sensors J.*, 4, 464–470, 2004.
- C. H. Wu, C. A. Zorman, M. Mehregany, "Fabrication and Testing of Bulk Micromachined Silicon Carbide Piezoresistive Pressure Sensors for High Temperature Applications" *IEEE Sensors Journal*, 6, 316, 2006.
- R. S. Okojie, A. R. Atwell, K. T. Kornegay, S. L. Roberson and A. Beliveau, 'Design considerations for bulk micromachining 6H–SiC high-G piezoresistive accelerometers' *Tech. Digest* 15th Int. Conf. on MEMS, 618–622, 2001.

- A. R. Atwell, R. S. Okojie, K. T. Kornegay, S. L. Roberson, and A. Beliveau, 'Simulation, fabrication and testing of bulk micromachined 6H-SiC high-g piezoresistive accelerometers' *Sensors and Actuators* A, Vol. 104, p. 11. 2003.
- A. P. Lee, A. P. Pisano and M. G. Lim, "Impact, Friction and Wear Testing of Microsamples of Polycrystalline Silicon," *Mat Res. Soc. Symp. Proc.*, 276, 67–78, 1992.
- S. Sundararajan and B. Bhushan, "Static friction and surface roughness studies of surface micromachined electrostatic micromotors using an atomic force/friction force microscope", *J. Vac. Sci. Technol.* A 19, 1777–1785, 2001.
- 66. K. Breuer, F. Ehrich, L. Frechette, S. Jacobson, C-C. Lin, D. Orr, E. Piekos, N. Savoulides, and C-W. Wong, "Challenges for Lubrication in High-Speed MEMS," in *NanoTribology*, Edited by S. Hsu., Kluwer Press, 2002.
- Sniegowski J J Tribology Issues and Opportunities in MEMS ed B Bushan (Netherlands:Kluwer–Academic), 325–331, 1998.
- 68. Krauss, A. R.; Auciello, O.; Gruen, D. M.; Jayatissa, A.; Sumant, A.; Tucek, J.; Mancini, D. C.; Moldovan, N.; Erdemir, A.; Ersoy, D.; Gardos, M. N.; Busmann, H. G.; Meyer, E. M.; Ding, M. Q., "Ultrananocrystalline diamond thin films for MEMS and moving mechanical assembly devices ", *Diamond and Related Materials*, **10** 1952–1961, 2001.
- W. R. Ashurst, M. B. J. Wijesundara, C. Carraro and R. Maboudian, "Tribological impact of SiC encapsulation of released polycrystalline silicon microstructures", *Tribology Letters*, 17, pp. 195–198, 2004.
- Flynn, A. M., Tavrow, L. S.; Bart, S. F.; Brooks, R. A.; Ehrlich, D. J.; Udayakumar, K. R.; Cross, L. E., "Piezoelectric micromotors for microrobots", *J. MEMS*, 1 44–51, 1992.
- Y. Yoshino, "Transversal Type Piezoelectric Resonator Using ZnO Thin Film on Micro-Fabricated Elinver (Fe-Ni-Cr-Ti) Alloy", *Mat. Res. Soc. Symp. Proc.*, **518**, 9–15, 1998.
- I. Lee, H. J. Sung, "Development of an array of pressure sensors with PVDF film", *Experiments in Fluids*, 26, 27–35, 1999.
- M. J., Lowe, "On the Integration of Piezoelectric Materials into MEMS Devices", *National Physics Laboratory Journal*, 15, 125–132, 2002
- H. Kahn, M. A. Huff, and A. H. Heuer, "The TiNi shapememory alloy and its applications for MEMS" *J. Micromech. Microeng.* 8, 213–221, 1998.
- Y. Fu, H. Du, W. Huang, S. Zhang and M. Hu, "TiNi-based thin films in MEMS applications: a review", *Sensors and Actuators A* 112, 395–408, 2004
- Jung, B. Y., et al; Thermolelectric Properties of P-type (BilxSbx)2Te3 Fabricated by Mechanical Alloying Process, Mats. Res. Soc. Symp. Proc., 478 (Thermoelectric materials-New Directions and Approaches) 121–126, 1997.
- K. Baert, P. Mertens and P. van Gerwen, "thin film poly-Si70%Ge30% for themopiles", Proc. Micro. System. Technologies'98, pp. 655–658.
- O. Paul, M. von Arx, H. Baltes, "Process-dependent thermophysical properties of CMOS IC thin films", Proceedings Transducers'95/Eurosensors XI, pp. 178–181.
- L. Jr. Ching, H. J. H. Chen, I. Y. Huang, and S. R. S. Huang, "A poly-Si thermoelectric cooling device fabricated by surface micromachining technology", IEEE Proc. 12th international conference on transducers, solid-state sensors, actuators, and microsystems, 2003, pp. 1084–1087.
- L. Ruiz de Angulo, J. S. Abell, and I. R. Harris, "Magnetostrictive properties of polymer bonded Terfenol-D", *Journal of Magnetism and Magnetic Materials*, 157–158, 508–509, 1996.
- N. J. Grabham, N. M. White and S. P. Beeby, "Thick-film magnetostrictive material for MEMS", *Electronics Letters*, 36, 332–334, 2000.
- 82. H. Becker, and C. Gärtner, "polymer microfabrication

methods for microfluidic ananlytical applications", *Electrophoresis* **21**, 12–26, 2000.

- E. M. Schmidt, J. S. McIntosh, and M. J. Bak, "Long-Term Implants of Parylene-C Coated Microelectrodes" *Medical* and Biological Engineering and Computing, 26 96–101, 1988.
- 84. G. M. Atkinson, R. E. Pearson, Z. Ounaies, C. Park, J. S. Harrison, W. C. Wilson, and J. A Midkiff, "Piezoelectric Polyimide MEMS Process" Proceedings of the NASA VLSI Symposium, May 2003.
- Y. Bar-Cohen, "Nanotechnology Using Electroactive Polymers as Artificial Muscles MSTnews, International Newsletter on Microsystems and MEMS, 3, 45–46, 2001.
- A. Husain, J. Hone, Henk W. Ch. Postma, X. M. H. Huang, T. Drake, M. Barbic, A. Scherer, and M. L. Roukes, "Nanowire-based very-high-frequency electromechanical resonator", *Appl. Phys. Lett.*, 83, 1240–1242, 2003.
- H. B. Peng, C. W. Chang, S. Aloni, T. D. Yuzvinsky, and A. Zett, "Ultrahigh Frequency Nanotube Resonators", *Phys. Rev. Lett.*, 97, 087203, 2006
- Y. Cui, Q. Wei, H. Park, C. M. Lieber, "Highly Sensitive and Selective Detection of Biological and Chemical Species", *Science* 293 1289, 2001.
- G. Zheng, F. Patolsky, Y. Cui, W. U. Wang, C. M. Lieber," Multiplexed electrical detection of cancer markers with nanowire sensor arrays", *Nature Biotech.* 23 1294–1301, 2005.
- E. S. Forzani, H. Zhang, L. A. Nagahara, I. Amlani, R. Tsui, N. Tao, "A Conducting Polymer Nanojunction Sensor for Glucose Detection", *Nano Lett.* 4 1785–1788, 2004.
- K. Ramanathan, M. A. Bangar, M. Yun, W. Chen, N. V. Myung, A. Mulchandani, "Bioaffinity Sensing Using Biologically Functionalized Conducting-Polymer Nanowire", *J. Am. Chem.* Soc. 127, 496–497, 2005.
- Z. L. Wang and J. H. Song, "Piezoelectric Nanogenerators Based on Zinc Oxide Nanowire Arrays", *Science*, **312**, 242–246, 2006.
- B. Ashrafi, P. Hubert and S. Vengallatore, "Carbon nanotube-reinforced composites as structural materials for microactuators in microelectromechanical systems" *Nanotechnology* 17, 4895–4903, 2006.
- 94. B. W. Chui, T. D. Stowe, Y. S. Ju, K. E. Goodson, T. W. Kenny, H. J. Mamin, B. D. Terris, R. P. Ried, and D. Rugar, "Low-Stiffness Silicon Cantilevers with Integrated Heaters and Piezoresistive Sensors for High-Density AFM Thermomechanical Data Storage", *JMEMS*, 7, 69–78, 1998.
- R. R. O'Keefe, R. W. Carpick, R. F. Cooper, W. J. Drugan, and D. S. Stone, "Oxide nanocomposites for MEMS applications," Proc. of the Society for Experimental Mechanics Annual Conference 160, 2000.
- T. D. Stowe, K. Yasumura, and T. W. Kenny, D. Botkin, K. Wago, and D. Rugar, "Attonewton force detection using ultrathin silicon cantilevers", *Applied Physics Letters*, 71, 288–290, 1997.
- Jinling Yang, Takahito Ono, and Masayoshi Esashi, "Energy Dissipation in Submicrometer Thick Single-Crystal Silicon Cantilevers", *JMEMS*, 11, pp. 775–783, 2002.
- Z. Lee, C. Ophus, L. M. Fischer, N. Nelson-Fitzpatrick, K. L. Westra, S. Evoy, V. Radmilovic, U. Dahmen and DMitlin, "Metallic NEMS components fabricated from nanocomposite Al–Mo films" *Nanotechnology* 17, 3063–3070, 2006.
- A. J. Auberton-Herve, "SOI: Materials to systems", IEEE Proc. of 1996 international electron devices meeting, pp. 3–10.
- 100. R. Simonton, "SOI wafer technology for CMOS ICs", http://www.icknowledge.com/threshold_ simonton/soitechnology.pdf
- 101. "IBM Research demonstrates path for extending current chipmaking technique", 2006, http://domino.watson.ibm.com/ comm/pr.nsf/pages/news.20060220_nemorelease.html

- 102. J. Brook and R. Dandliker, "Submicrometer holographic photolithography", Solid State Technol. 32, pp. 91–94, 1989.
- B. Omar, S. Clube, F. Hamidi, M. D. Struchen, and S. Gray, "Advances in holographic lithography", Solid State Technol. 34, pp. 89–94, 1991.
- 104. K. Ikuta, K. Hirowatari, and T. Ogata, "Three-dimensional integrated fluid systems (MIFS) fabricated by stereo lithography", IEEE proc. of international workshop on Micro Electro Mechanical Systems (MEMS' 94). pp. 1–6.
- 105. C. M. Waits, R. Ghodssi, M. H. Ervin, and M. Dubey, "MEMS-based gray-scale lithography", IEEE proc. of 2001 international semiconductor device research symposium, pp. 182–185.
- 106. S. C. Jacobsen, D. L. Wells, C. C. Davis and J. E. Wood, "Fabrication of Micro-Structures Using Non-Planar Lithography", IEEE proc. of MEMS'91. pp. 45–50.
- 107. R. J. Jackman, J. L. Wilbur, and G. M. Whitesides, "Fabrication of Submicrometer Features on Curved Substrates by Microcontact Printing", *Science*, 269, 664–666, 1995.
- 108. W. J. Li, J. D. Mai, and C. M. Ho, "A MEMS fabrication technique for non-planar substrates", IEEE proc. of the 11th annual international workshop on MEMS, 1998, pp. 268–273.
- 109. G. J. Burger, E. J. T. Smulders, J. W. Berenschot, T. S. J. Lammerink, J. H. J. Fluitman and S. Imai, "High Resolution Shadow Mask Patterning in Deep Holes and its Application to an Electrical Wafer Feed-Through", Proc. The 8th International Conference on Solid-State Sensors and Actuators, and Eurosensors IX. 573–576, 1995.
- 110. M. Y. Jung, W. I. Jang, C. A. Choi, M. R. Lee, C. H. Jun, Y. T. Kim, "Novel lithography process for extreme deep trench by using laminated negative dry film resist", IEEE proc. of 17th international conference on MEMS, pp. 685–688, 2004.
- 111. P. B. Chu, J.T. Chen, R. Yeht, G. Lin, J. C. P. Huang, E. A. Warneket, and K. S. J. Pister, "Controlled Pulse-Etching with Xenon Difluoride", Proc. of 1997 International Conference on Solid-state Sensors and Actuators, Chicago, June 16–19, 1997, pp. 665–668
- 112. Laermer F, Schilp A, Funk K, Offenberg M, "Bosch deep silion etching: improving uniformity and etch rate for advanced MEMS applications" *Technical Digest MEMS*'99, *Florida*, USA, 1999, pp. 211–16.
- 113. L. Jiang, R. Cheung, R. Brown, A. Mount, "Inductively coupled plasma etching of SiC in SF₆/O₂ and etch-induced surface chemical bonding modifications", *J. Appl. Phys.* 93, 1376, 2003.
- 114. L. Jiang, N. Plank, R. Cheung, M. A. Blauw and E. van der Drift, "Dry etching of SiC in inductively coupled Cl₂/Ar plasma", *J. of Phys. D: Applied Physics*, **37** 1809–1816, 2004.
- 115. Guckel H, Skrobis K J, Christension T R, Klein J, Han S, Choi B and Lovel E G, "Fabrication of assembled micromechanical components via deep x-ray lithography" Proc. MEMS'91 (IEEE, Nara, Japan), 74–79, 1991.
- 116. HarmeningM, BacherW, Bley P, Kholi A, Kalb H, Kowanz B, Michel A and Mohr K, "Molding of threedimensionalmicrostructures by the LIGA process" Proc. MEMS'92 (IEEE, Travem"unde, Germany), 202–207, 1992.
- 117. Shew B Y, Cheng Y, Lin C H, Ma W P, Huang G J, Kuo C L, Tseng S C, Lee D S and Chang G L, "Manufacturing process for LIGA spinnerets", *Sensors and Materials* 11, 329–337, 1999.
- 118. A. A. Yasseen, C. H. Wu, and C. A. Zorman, "Fabrication and Testing of Surface Micromachined Polycrystalline SiC Micromotors", IEEE Electron. Dev. Lett., 21, 164–166, 2000.
- 119. J. Goettert; Y.M. Desta, B. Loechel, "Commercialization of Direct LIGA - Challenges and Opportunities", Proc. COMS2004, Edmonton, Aug 2004.
- H. Guckel, "High-aspect-ratio micromachining via deep x-ray lithography", Proc. IEEE, 86, pp. 1586–1593, 1998.

- 121. Lasky J B, "Wafer bonding for silicon-on-insulator technologies", Appl. Phys. Lett. 48 78-80, 1986.
- 122. K. Petersen, J. Brown, T. Vermeulen, P. Barth, J. Mallon and J. Bryzek, "Ultra-stable, high-temperature pressure sensors using silicon fusion bonding", *Sensors and Actuators A*, 21, 96–101, 1990.
- 123. J. J. Lerou, M. P. Harold, J. Ryley, J. Ashmead, T. C. O'Brien, M. Johnson, J. Perrotto, C. T. Blaisdell, T. A. Rensi, and J. Nyquist, "Microsystem technology for chemical and biological microreactors," *DECHEMA Monographs*, vol. 132, no. 51, 1996.
- 124. N. Miki, X. Zhang, R. Khanna, A. A. Ayón, D. Ward and S. M. Spearing "Multi-Stack Silicon-Direct Wafer Bonding for 3D MEMS Manufacturing" *Sensors and Actuators*, Part A, Physical, **103** (1–2): 194–201, 2003.
- 125. T. R. Anthony, "Anodic bonding of imperfect surfaces" J. Appl. Phys. 54, 2419–2428, 1983.
- 126. H. Wada, H. Sasaki and T. Kamijoh, "Wafer bonding technology for optoelectronic integrated devices", Solid-State Electronics, 43, pp. 1655–1663, 1999.
- 127. U. Gösele and Q. Y. Tong, "Semiconductor wafer bonding", Annu. Rev. Mater. Sci. 28, 215–241, 1998.
- 128. Q.Y. Tong, and U. Gösele, "Semiconductor Wafer Bonding: Science and Technology", John Wiley & Sons Inc., 1998.
- 129. S. S. Iver, and A. J. Auberton-Herve, "Silicon Wafer Bonding Technology for VLSI and MEMS Applications", Institution of Engineering and Technology, 2001.
- M. A. Schmidt, "Silicon wafer bonding for micromechanical devices," Proc. Solid-State Sensor and Actuator Workshop at Hilton Head, 1994.
- 131. C. Gui, M. Elwenspoek, J. G. E. Gardeniers, and P. V. Lambeck, "Present and Future Role of Chemical Mechanical Polishing in Wafer Bonding", J. Electrochem. Soc., 145, pp. 2198–2204, 1998.
- 132. A. R. Mirza, "One Micron Precision, Wafer-Level aligned bonding for interconnect, MEMS and packaging applications", *IEEE proceedings of Electronic Components & Technology Conference*, 676–80, 2000.
- 133. Lu J Q, Kwon Y, Rajagopalan G, Gupta M, McMahon J, Lee K W, Kraft R P, McDonald J F, Cale T S, Gutmann R J, Xu B, Eisenbraun E, Castracane J, Kaloyeros A, "A wafer-scale 3D IC technology platform using dielectric bonding glues and copper damascene patterned inter-wafer interconnects", Proceedings of the IEEE 2002 International Interconnect Technology Conference, 2002, 78–80.
- 134. Liudi Jiang, G Pandraud, P J French, S. M. Spearing and M. Kraft, "A novel method for nanoprecision alignment in wafer bonding applications", *J. Micromech. and Microeng.*, 17, S61–S67, 2007.
- 135. V. Dragoi, S. Farrens, P. Lindner and J. Weixlberger, "Low temperature wafer bonding for microsystems applications", IEEE proc. of 2004 international semiconductor conference, 1 (2004) pp. 199–202.
- 136. A. Sanz-Velasco, P. Amirfeiz, S. Bengtsson, and C. Colinge, "Room Temperature Wafer Bonding Using Oxygen Plasma Treatment in Reactive Ion Etchers With and Without Inductively Coupled Plasma", J. Electrochem. Soc., 150 (2003) pp. G155–G162.
- 137. R. Darveaux, and L. Munukutla, "Critical challenges in packaging MEMS devices", IEEE proc. of 2005 advanced semiconductor manufacturing conference and workshop, pp. 210–216.
- 138. M. da Silva, "Standard Open Tool Packages for MEMS enabled products," Advanced Packaging Magazine, 2004.
- 139. A. R. Mirza, "Wafer-level packaging technology for MEMS", IEEE proc. of the 17th intersociety conference on thermal and thermomechanical phenomena in electronic systems, 2000, pp. 113–119.

- 140. M. Moraia and M. Amiotti, "Getters films at wafer level for wafer to wafer bonded MEMS" IEEE proc. of symposium on design, test, integration and packaging of MEMS/MOEMS 2003, pp. 346–349.
- 141. M. A. Haque, M. T. A. Saif, "A review of MEMS-based microscale and nanoscale tensile and bending testing" Exp. Mech. 43, 248–255, 2003.
- 142. V. T. Srikar and S. M. Spearing. "Mechanical Test Methods for MEMS Devices" *Experimental Mechanics* 43 (3) 228–237, 2003.
- 143. H. D. Espinosa, B. C. Prorok, M. A. Fischer, "Methodology for determining mechanical properties of freestanding thin films and MEMS materials" *J. Mech. Phys. Sol.* 51, 47–67, 2003.
- 144. M-TEST: A test chip for MEMS material property measurement using electrostatically actuated testStructures, Osterberg, P. M., Senturia, S. D., *J. Microelectromechanical Systems*, **6**, 107–118, 1997.
- 145. Fracture testing of bulk silicon microcantilever beams subjected to a side load, Wilson, C. J., Beck, P. A., *J. Microelectromechanical Systems*, **5**, 142–150, 1996.
- 146. A New Bulge Test Technique for the Determination of Young's Modulus and Poisson's Ratio of Thin Films, Vlassak, J.J., Nix, W.D., J. Mat. Res., 7, 3242–3249, 1992.
- 147. V. T. Srikar, A. K. Swan, M. S. Unlu, B. B. Goldberg and S. M. Spearing. "Raman Spectroscopy of Deep Reactive Ion Etched Silicon Flexures: Microscale Stress Measurement and Model Verification" *J. MEMS*, **12**, 779–787, 2003.
- 148. Sharpe WN, Bagdahn J, Jackson K, Coles G Tensile testing of MEMS materials—recent progress, J. Mat. Sci., **38**, 4075–4079, 2003.
- 149. Mechanical Property Measurement of 0.5μm CMOS Microstructures, Lu, M. S-C., Zhu, X., Fedder, G. K., Mat. Res. Soc. Symp. Proc., 518, 27–32, 1998.
- 150. J. J. Vlassak, W. D. Nix, A New Bulge Test Technique for the Determination of Young's Modulus and Poisson's Ratio of Thin Films, J. Mat. Res., 7, 3242–3249, 1992.
- 151. C. S. Oh and W. N. Sharpe, Techniques for measuring thermal expansion and creep of polysilicon, Sensors and Actuators A-Physical **112**, 66–73, 2004.
- 152. Round-Robin Tests of Modulus and Strength of Polysilicon, Sharpe, W. N., Brown, S., Johnson, G. C., Knauss, W., Mat. Res. Soc. Symp. Proc., 518, 57–65, 1998.
- 153. W. N. Sharpe, "Murray lecture—Tensile testing at the micrometer scale: Opportunities in experimental mechanics", *Exp. Mech* 43, 228–237, 2003.
- 154. G. J. McShane, M. Boutchich, A. S. Phani, D. F. Moore, T. J. Lu, Young's modulus measurement of thin-film materials using micro-cantilevers, J. Micromechanics and Microengineering, 16, 1926–1934, 2006.
- 155. W. D. Nix, Elastic and plastic properties of thin films on substrates: nanoindentation techniques, Materials Science and Engineering A. 234, 37–44, 1997.
- 156. H. Kahn, N. Tayebi, R. Ballarini, R. L. Mullen, A. H. Heuer, Fracture toughness of polysilicon MEMS devices, Sensors and Actuators, A-Physical, 82, 274–280, 2000.
- 157. K. S. Chen, A. Ayon and S. M. Spearing, "Controlling and testing the fracture strength of silicon on the mesoscale", J. Am. Ceram. Soc. 83, 1476–1484, 2000.
- 158. C. J. Wilson, P. A. Beck, Fracture testing of bulk silicon microcantilever beams subjected to a side load, J. Microelectromechanical Systems, 5, 142–150, 1996.
- 159. K. S. Chen, S. M. Spearing, N. N. Nemeth, "Structural design of a silicon micro-turbo-generator", *AIAA J.*, **39** (4): 720–728, 2001.
- 160. M. A. Huff, A. D. Nikolich and M. A. Schmidt, "Design of Sealed Cavity Microstructures Formed by Silicon Wafer Bonding", J. MEMS, 2, 74–81, 1993.

- 161. D. S. Walters and S. M. Spearing, "On the Flexural Creep of Single Crystal Silicon", *Scripta Materialia*. 42, 769–774, 2000.
- 162. H. S. Moon, D. Choi and S. M. Spearing "Development of Si/SiC Hybrid Structures for Elevated Temperature Micro-Turbomachinery" J. MEMS 13(4) 676–687, 2004.
- 163. Analysis of Critical Debonding Pressures of Stressed Thin-Films in the Blister Test, Allen, M. G., Senturia, S.D., J. Adhesion, 25, 303–315, 1988.
- 164. A new Procedure for Measuring the Decohesion Energy for Thin Ductile Films on Substrates, Bagchi, A., Lucas, G. E., Suo, Z., Evans, A. G., J. Mat. Res., 9, 1734–1741, 1994.
- 165. M. P. De Boer, J. A. Knapp, T. A. Michalske, U. Srinivasan, R. Maboudian Adhesion hysteresis of silane coated microcantilevers, *Acta Mater.* 48, 4531–4541, 2000.
- 166. Q-Y Tong and U. Gosele "Semiconductor Wafer Bonding: Science and Technology" Wiley, New York, 1999.
- 167. W. P. Maszara, G. Goetz, A. Caviglia, J. B. McKitterick, "Bonding of Silicon Wafers for Silicon-on-Insulator" J. Appl. Phys., 64, 4943–4950, 1988.
- 168. J. Bagdahn, M. Petzold, A. Ploeβl and M. Wiemer, "Measurement of the Local Strength Distribution of Directly Bonded Silicon Wafers Using the Micro-Chevron Test" Electrochemical Society Proceedings, 99, 218–223, 2001.
- 169. K. T. Turner, S. M. Spearing, "Mechanics of direct wafer bonding" *Proc. Roy. Soc. Lond. A.* 462, 171–188, 2006.
- 170. M. G. Allen, M. Mehregany, R. T. Howe, S. D. Senturia, "Microfabricated Structures for the In Situ Measurement of Residual Stress, Young's Modulus and Ultimate Strain of Thin-Films," *Appl. Phys. Letters*, **51**, 241–243, 1987.
- 171. Zhang, X., Zhang, T.Y., Zohar, Y., Measurements of residual stresses in thin films using micro-rotating-structures, *Thin Solid Films*, **335**, 97–105, 1998,
- 172. Hommel, M., Kraft, O., Arzt, E., Baker, S. P., Micro-Tensile and Fatigue Testing of Copper Thin Films on Substrates, *Mat. Res. Soc. Symp. Proc.*, **546**, 133–139, 1999.
- 173. G. Cornella, R. P. Vinci, R.Suryanarayanan Iyer, R. H. Dauskardt, Observations of Low Cycle Fatigue of Al Thin Films for MEMS applications, *Mat. Res. Soc. Symp. Proc.*, 518, Materials Research Society, 81–86, 1998.
- 174. C. L. Muhlstein, and S. B. Brown, Reliability and Fatigue Testing of MEMS, in Tribology Issues and Opportunities in MEMS, Ed. Bhusan, B., Kluwer, 529–537, 1998.
- 175. H. Kahn, L. Chen, R. Ballarini, A. H. Heuer, "Mechanical fatigue of polysilicon: Effects of mean stress and stress amplitude" *Acta Mater.* 54, 667–678, 2006.
- 176. D. H. Alsem, O. N. Pierron, E. A. Stach, C. L. Muhlstein, R. O. Ritchie, "Mechanisms for fatigue of micron-scale silicon structural films", *Adv. Eng. Mats.* 9, 15–30, 2007.
- 177. MEMS Opportunities in Accelerometers and Gyros and the Microtribology Problems Limiting Commercialization, Souloff, R. E. in Tribology Issues and Opportunities in MEMS, Proceedings of the NSF/AFOSR/ASME Workshop on Tribology Issues and Opportunities in MEMS held in Columbus, Ohio, USA 9-11 November 1997. Edited by Bhusan, B. Kluwer. Dordrecht/Boston/London 1998.
- 178. Surface-micromachined gear trains driven by an on-chip electrostatic microengine, Sniegowski, J. J., Garcia, E. J., *IEEE Electronic Device Letters*, **17**, 366–368, 1996.
- 179. J. A. Williams and H. R. Le, "Tribology and MEMS", *J. Phys D*, **39**, R201–R214, 2006.
- 180. K. Komvopolous, Micromechanics and Surface Modification Methods for Reducing Adhesion in Microelectromechanical Systems, Mat. Res. Symp. Proc. 546, 145–151, 1999.
- 181. S-H. Lee, M-J. Kwon, J-G. Park, Y-K. Kim, H-J. Shin, "The Surface Modification with Fluorocarbon Thin Films for the Prevention of Stiction in MEMS", Mat. Res. Soc. Symp. Proc. 518, 143–148, 1998.
- 182. L. X. Liu, Z. S. Spakovszky, "Effects of bearing stiffness

anisotropy on hydrostatic micro gas journal bearing dynamic behavior", *J. Eng. Gas Turbines and Power, Trans. ASME*, **129** (1): 177–184, 2007.

Ċ

Liudi Jiang received her BSc. in Microelectronics from Hefei University of Technology and MSc. from the Institute of Solid State Physics, Chinese Academy of Sciences, both in P.R. China and her Ph. D in 2001 from the University of Dundee. She conducted postdoctoral research at The Scottish Microelectronics Centre,

University of Edinburgh from 2001 to 2005 where she was a key contributor towards the development of state-of-the-art SiC microelectromechanical systems (MEMS) for applications in harsh environments. In May 2005, Dr. Jiang joined the Materials Research Group at the School of Engineering Sciences (SES), Southampton University as a Postdoctoral Researcher and in 2006, she was appointed as the Research Council UK (Roberts) Academic Fellow in Nanomaterials hosted by the SES. Her research activities have so far led to over 30 publications and two patent applications. Dr. Jiang's research interest covers a range of areas in nanomaterials and nanomanufacturing. Her research focuses on the development of advanced nanomaterials with a view to explore their applications in nano/micro engineering systems. She also works on developing innovative nano/micro processes for the next generation nanoelectronic devices and MEMS/NEMS.



S. Mark Spearing received the Ph.D degree from Cambridge University Engineering Department in 1990. In 2004 he was appointed Professor of Engineering Materials in the School of Engineering Sciences at The University of Southampton, UK. Prior to this he spent ten years as Professor of Aeronautics and Astronautics at the

Massachusetts Institute of Technology (MIT), Cambridge, USA. His technical interests include materials characterization and structural analysis and design of MEMS, development of wafer bonding-technologies, micro-electronic and MEMS packaging and advanced composites. From 1995 to 2005 he was responsible for materials, structural design and packaging tasks of the MIT MicroEngine, MicroRocket, Micro-Chemical Power and MicroHydraulic Transducer projects as well as conducting crosscutting underpinning technology development. He is an editor of the Journal of Microelectromechanical Systems and Member of ASME. In 2004 he received a Royal Society Wolfson Research Merit Award.